Author's response to reviews

Title: Binge-eating disorder diagnosis and treatment. A recap in front of DSM-5.

Authors:

   Federico Amianto (federico.amianto@unito.it)
   Luisa Ottone (luisa.ottone@gmail.com)
   Giovanni Abbate Daga (giovanni.abbatedaga@unito.it)
   Secondo Fassino (secondo.fassino@unito.it)

Version: 5 Date: 18 February 2015

Author's response to reviews:

Dear Referees we kindly thank you for your revisions which improved our manuscript a lot.

In the last revision we replaced with the definition of "evidence based" the previous "promising" attribute to IPT. We also arranged the spelling of "treatment" and "BWL" as indicated.

Thank you very much and best regards.